Author's response to reviews

Title: Intima-media thickness and arterial function in obese and non-obese children

Authors:

Heidi Weberruss (heidi.weberruss@tum.de)
Raphael Pirzer (raphael.pirzer@med.uni-wh.de)
Birgit Böhm (birgit.boehm@tum.de)
Robert Dalla Pozza (robert.dallapozza@med.uni-muenchen.de)
Heinrich Netz (heinrich.netz@med.uni-muenchen.de)
Renate Oberhoffer (renate.oberhoffer@tum.de)

Version: 2
Date: 29 April 2015

Author's response to reviews:

• Email addresses of all authors was included on the title page
• Ethics Committee was specified (Ethics Committee of the Technische Universität München) in the method section
• A competing interest section was added
• An acknowledgments section was added